Author's response to reviews

**Title:** Deletion of the TNFAIP3/A20 gene detected by FICTION analysis in classical Hodgkin lymphoma

**Authors:**

Junko Nomoto (jnomoto@ncc.go.jp)
Nobuhiro Hiramoto (nccnsct@gmail.com)
Motohiro Kato (katom-tky@umin.ac.jp)
Masashi Sanada (sanada-tky@umin.net)
Akiko M Maeshima (akmaeshi@ncc.go.jp)
Hirokazu Taniguchi (hitanigu@ncc.go.jp)
Fumie Hosoda (fhosoda@ncc.go.jp)
Yoshitaka Asakura (yasakura@ncc.go.jp)
Wataru Munakata (wmunakat@ncc.go.jp)
Dai Maruyama (dmaruyam@ncc.go.jp)
Takashi Watanabe (takawata@ncc.go.jp)
Hitoshi Nakagama (hnakagam@ncc.go.jp)
Kengo Takeuchi (kentakeuchi-tky@umin.net)
Kensei Tobinai (ktobinai@ncc.go.jp)
Seishi Ogawa (sogawa-tky@umin.ac.jp)
Yukio Kobayashi (ykkobaya@ncc.go.jp)

**Version:** 3  **Date:** 6 September 2012

**Author's response to reviews:** see over
September 6th, 2012

Editorial Office, BMC Cancer

Dear Editor:

Please find enclosed our revised manuscript entitled “Deletion of the TNFAIP3/A20 gene detected by FICTION analysis in classical Hodgkin lymphoma” by Junko Nomoto, et al., which we are submitting for consideration of publication in the BMC Cancer as an Original Article. Please find the attached clear and tracked copy of the manuscript.

We believe that all the revisions were made according to the reviewers’ comments. We have asked Edanz Editing to edit our paper, with refinement and improvement of the previous version.

This manuscript is not under consideration for publication elsewhere, and all authors have directly participated in the present study and have approved the final version. None of the authors has any significant financial relationships to disclose.

We hope that our manuscript will be suitable for publication in the BMC Cancer.

Sincerely yours,

Yukio Kobayashi, MD, PhD
Hematology and Stem Cell Transplantation Division,
National Cancer Center Hospital,
5-1-1 Tsukiji, Chuo-ku,
Tokyo 104-0045, Japan
Phone: +81-3-3542-2511
FAX: +81-3-3542-3815
E-mail: ykkobaya@ncc.go.jp